Author’s response to reviews

Title: A rare case of recurrent primary anorectal melanoma emphasizing the importance of postoperative follow-ups

Authors:

Zhihao Li (zhihaoli91@gmail.com)

Peter Sandera (peter.sandera@triemli.zuerich.ch)

Markus Weber (markus.weber@triemli.zuerich.ch)

Version: 2 Date: 28 Feb 2020

Author’s response to reviews:

We thank the reviewers for their comments, according to which the manuscript has been revised. In addition, we added the histologic pictures which was requested from Reviewer 2 last time in the Figure 3.